Author’s response to reviews

Title: Impact of the Family Health Program on the quality of vital information and reduction of child unattended deaths in Brazil: an ecological longitudinal study

Authors:

Davide Rasella (davide.rasella@gmail.com)
Rosana Aquino (aquino@ufba.br)
Mauricio L Barreto (mauricio@ufba.br)

Version: 3 Date: 11 June 2010

Author’s response to reviews:

Dear Dr Natalie Pafitis

Thanks for your letter and your acceptance of the manuscript conditional to the documentation of the name of the ethics committee which approved our study.

As we have added in the last sentence of the methods section: “no ethics committee approval was needed for our study, being the data used in this research obtained from a public-use data set.”

We have also checked that our manuscript conforms to the journal style.

Yours Sincerely

Davide Rasella